ECONOMIC PAPERS

COMMISSION OF THE EUROPEAN COMMUNITIES DIRECTORATE-GENERAL FOR ECONOMIC AND FINANCIAL AFFAIRS

Number 88

December 1991

The Adequacy and Allocation of World Savings

> Javier Santillán* Internal Paper



"Economic Papers" are written by the Staff of the Directorate-General for Economic and Financial Affairs, or by experts working in association with them. The "Papers" are intended to increase awareness of the technical work being done by the staff and to seek comments and suggestions for further analyses. They may not be quoted without authorisation. Views expressed represent exclusively the positions of the author and do not necessarily correspond with those of the Commission of the European Communities. Comments and enquiries should be addressed to:

The Directorate-General for Economic and Financial Affairs, Commission of the European Communities, 200, rue de la Loi 1049 Brussels, Belgium

ECONOMIC PAPERS

Number 88

December 1991

The Adequacy and Allocation of World Savings

Javier Santillán* Internal Paper

* J. Santillán is an economist in the Research Department of the Bank of Spain who was officially seconded to DG II of the European Commission. The author wishes to gratefully acknowledge Flemming Larsen for his helpful comments and suggestions which have certainly contributed to improve earlier versions of this paper. Useful remarks from other colleagues of the international monetary division of DG II are also acknowledged, and particularly those by Barbara Kauffmann. CONTENTS

Summary

List of tables

List of graphs

PAGE

1.	Overview of the paper	1
2.	Saving Trends and Possible Influences	2
3.	Adequacy of Savings: Some Theoretical Perspectives and Evidence	13
4.	Origin versus Allocation of Saving	21
5.	Current Account Balances and Capital Flows	26
6.	The European Community's Saving Performance in the Context of Economic and Monetary Union	33
7.	Conclusions	35
	Annex: Statistical Issues on Saving and Current Account	38
	References	41

SUMMARY

This paper reviews recent trends in savings in the industrial countries and discusses the factors that have influenced saving developments. It examines whether savings are adequate applying a number of theoretical approaches. The note emphasises the importance of distinguishing between origin and allocation of savings and highlights the relationship between current account balances and capital flows.

The paper argues that during the past decade, increasingly closely integrated financial markets imply that the adequacy and allocation of saving across countries have taken on a crucial significance for the international monetary system. By the same token, the pattern of saving and investment that will develop across the European Community will substantially determine whether the full benefits of Economic and Monetary Union will be reaped.

The paper concludes that multilateral surveillance should increasingly focus on the adequacy of savings across countries. This conclusion is particularly relevant for the final stage of EMU, when the introduction of a common currency eliminates the current account as an intermediate policy target.

TABLES

- 1. Average national saving in major industrial countries
- 2. Projected increases in dependency ratios of seven industrial countries
- 3. Medium-term effects of increased saving in industrial countries (percentage points variations respect to IMF World Economic Outlook baseline projections, period 1991-96)

GRAPHS

- 1. Net national savings in industrialised countries (% of GDP)
- 2. Real long-term interest rates in EC, USA and Japan (yearly averages)
- 3. Expansion of international financial activity (international bank lending, USD billions)
- 4. Current account balances of major industrial countries (% of GDP)
- 5. Current account imbalances of major industrial countries (weighted average of the sum of deficits and surpluses of G-7 countries as a % of GDP)

THE ADEQUACY AND ALLOCATION OF WORLD SAVINGS

1. OVERVIEW OF THE PAPER

The issue of the adequacy of world savings has been a focus of attention in the economic policy debate in recent years. For the decade of the 1990s the saving performance of the industrial countries, and the widely diverging patterns observed among them, give rise to new concerns due to the prospective increase in the demand for financial resources over the medium-term. The objective of this paper is to provide a comprehensive synthesis of recent research in this area and to draw some conclusions for economic policy.

in section 2 a short description of recent saving patterns is presented together with a summary of the most prominent explanations for those patterns and the implications for policies aimed at fostering higher savings.

Section 3 analyzes the key issue of the adequacy of savings. The question is addressed in light of the evidence that can be obtained from growth models. The conclusions obtained generally confirm that the present level of savings in industrial countries is suboptimal, notably in the United States. In this latter case, simulations based on growth models highlight the serious negative consequences of low saving for the US economy, even if it does not pose problems of sustainability. Section 4 discusses the degree of international capital mobility reached in recent years, as a key element to be considered with regard to the adequacy and international allocation of savings.

Section 5 explores the extent to which the causality between current account imbalances and capital flows may have changed in recent years. Financial developments which have affected the composition of capital flows, are briefly analyzed, and the consequences and implications for macroeconomic policy coordination are considered. Section 6 briefly examines saving developments in the European Community and their implications for Economic and Monetary Union. The main conclusions are summarized in section 7.

2. SAVING TRENDS AND POSSIBLE INFLUENCES

Following a fairly stable situation during the 1960s and 1970s the share of national income saved in the industrial countries fell significantly in the 1980s. As a percent of GDP, the average saving performance during the most recent decade in net terms (i.e., allowing for depreciation of the capital stock), was 3.5 percentage points below that of the period 1960-80, (see Table 1 and Chart 1).

TABLE 1

_ 3 _

AVERAGE NATIONAL SAVING IN MAJOR INDUSTRIAL COUNTRIES

Countries and periods	Gross National Saving(1)	Net National Saving(2)	Of which:			
		g(-)	Public		Private	
				Total	Households	Bus iness
United States						
1960-69	19.7	10.8	0.8	10.0	6.2	3.8
1970-79	19.4	9.1	-1.2	10.3	7.6	2.6
1980-89	16.3	4.0	-3.8	7.8	6.0	1.9
Japan						
1960-69	34.5	25.2	6.6	18.6	11.9	6.6
1970- 79	35.3	25.6	5.0	20.6	16.5	4.1
1980-89	31.6	20.9	5.1	15.7	13.1	2.6
<u>Germany</u>						
1960-69	27.3	19.9	6.3	13.5	7.6	6.0
1970-79	24.3	15.2	3.7	11.5	9.7	1.7
1980-89	22.5	11.6	1.5	10.1	8.9	1.2
<u>Seven Major</u>						
Indus. countri	<u>les(</u> 3)					
1960-69	22.0	13.5	2.3	11.2	7.2	4.0
1970-79	23.2	13.5	0.8	12.7	10.0	2.7
1980-89	21.5	10.0	-0.9	10.9	8.8	2.1

Source: BIS annual report 1991 and OECD data

(1) As a percentage of GDP

2)

(2) As a percentage of national income

(3) Weighted average applying exchange rates of 1963, 1975 and 1988, respectively, for each of the three periods. As a consequence of the appreciation of the yen in the 1980s, the magnitude of the decrease in aggregate savings in that decade is somewhat understated.





Source: Commission Services

An important feature of developments during the 1980s is the fact that both public and private saving fell markedly. This contrasts with the experience of the 1970s when a sharp fall in public sector savings was more than offset by an increase in private savings¹). At the same time, differences in saving performances among the major industrial countries widened significantly as the progressive liberalisation of capital movements promoted a sharp expansion of both portfolio and direct investment which helped to accommodate the decline in savings in some countries, notably the United States. The combined effect of the decline in savings and the liberalisation of financial markets was a marked increase in inflation-adjusted interest rate in all countries (Chart 2).

The concerns caused by these developments have been amplified by the process of reforms in Eastern Europe and German unification which have increased pressures on the existing pool of saving. Taking into account also the investment needs associated with the reconstruction after the Gulf war, and the demand for resources in the developing world make it clear that the saving issue will continue to receive considerable attention during the 1990s.

An important subject in the discussion of saving levels and trends is the accounting criteria used to measure it. Divergence of national account rules among industrial countries must be considered when analysing and comparing trends. This issue is briefly addressed in the annex, together with the current account discrepancy problem.



GRAPH 2 - REAL LONG-TERM INTEREST RATES IN EC, USA AND JAPAN (Yearly averages)

Source: Commission Services

in attempting to understand the reasons for the decline in savings in the 1980s it is necessary to start with a clear view of what determines savings. The most widely accepted approach to the explanation of savings behaviour is based on the life-cycle hypothesis (Ando and Modigilani, 1963)²⁾, according to which saving decisions are a residual of consumption decisions, and the element which determines them is the aim of smoothing consumption over time. Accordingly, inflation, personal income, stock of wealth and demographic factors (such as life expectancy, age of retirement, age structure and family size) are major determinants of private saving.

Many studies testing the validity of life-cycle models in explaining saving patterns have been carried out in recent years³⁾. While the results obtained do not provide an unambiguous answer about the suitability of the models, it is widely accepted that variables usually included in life-cycle models do explain major trends and long-term fluctuations in savings. With regard to short-term fluctuations in savings, the explanatory power of these models seems more limited. However, if account is taken of liquidity constraints and of the

- 7 --

²⁾ The discussion in this context can include public savings performance. From the point of view of consumption smoothing, it is argued that government's behaviour should aim at keeping stable taxes when faced with the need to vary expenditure (see Barro, (1989). The following discussion refers to private savings.

³⁾ Comprehensive reviews are found in Aghevii et al. (1989); Smith (1989); and Dean et al. (OECD, 1990).

possibility of erroneous expectations on income evolution, the lifecycle framework remains a useful one⁴).

The influence of inflation on measured saving is twofold: by affecting personal income, and through the capital losses which inflation inflicts on holders of certain assets (like bonds or bank debosits). Even if such losses are compensated through increases in market interest rates inflation-adjusted household saving ratios may differ significantly from non-adjusted ratios, particularly during periods of accelerating or decelerating inflation. The evolution of inflation is therefore likely to have a substnatial impact on measured saving ratios (see annexed box for some references on the issue of measuring saving).

Demographic factors are considered as the main single factor behind broad trends in private savings over time and, to a great extent, contemporary differences among industrial countries⁵). Consequently, projections of future private saving trends largely rely on expected demographic developments. The "dependency ratio" (see Table 3) plays an essential role in this aspect, and its projected evolution explains long-run prospects for private saving behaviour. Indeed, projected

⁴⁾ A synthetic review of alternative approaches can be found in Green (1990). The common case of the so-called funconventional savings theories" is the rejection of the neoclassical paradigm on which the life-cycle model is based. "Unconventional" theories usually stress the significance of institutional factors in determining private saving trends. In particular corporate attitude towards savings, social norms, and demonstration effects among others are factors affecting private savings that are excluded from "conventional" models.

⁵⁾ See, for instance, Dean et al. (1990), Kauffmann (1990) or Crabam (1987). Sinn (1990, p.12) highlights the links between demography and net external assets in the 1970s and 1980s for Japan, Germany and other DECD countries. For a review of the macroeconomic effects of ageing population, see INF (1990).

increases in dependency ratios in the major industrial countries over the next 2 or 3 decades have given rise to fears of a continued deterioration in national saving performances.

The implementation of private pension schemes, is recognised as one of the main policy instruments able to promote household saving. This is owing to the effects of pension funds on the average retirement age (which is anticipated), to the low liquidity of pension benefits and to a misperception by fund's contributors that contributions to pension schemes constitue a tax rather than savings. Estimates on the effects of private pension funds have usually found significant positive effects on aggregate household savings of the increase of such funds.

Wealth and income distribution are identified as two key factors determining private saving behaviour in recent years. The sharp increase in household financial wealth during the 1980s in the industrial countries explains to a large extent the fail in saving propensity⁶⁾. Furthermore, changes in income distribution, combined fall. with demographic factors, have contributed to that In particular, the increased proportion of social security transfers income in total households' income tends to raise the marginal propensity to consume. Some estimates suggest that, in the United states, the consumption propensity out of labour income is 0.65 while the consumption propensity out of transfer income is 0.9 (Wilcox, 1991).

⁽³⁾ Wilcox (1991) reports that, for the United States, the rise in bousehold wealth between the mid-1970s and 1987 would explain, all sise equal, a decline of 3.5 percentage points in the private saving ratio. Likewise, the private saving propensity seems to have responded to the subsequent stock market adjustments.

A number of studies suggest that the response of savings to changes in interest rates is unclear even with regard to the direction of its effect. In this respect, theory cannot determine the sign of the elasticity of saving to interest rates, given the presence of offseting income and substitution effects. Whilst intuitively the preeminence of the substitution over the income effect could be expected (i.e. a positive interest rate-saving elasticity) empirical work has found widely diverging results, ranging from significant positive elasticities to very small ones or even none (see Smith, 1989).

The influence of taxation appears significant, and has been often identified as a factor discouraging savings in industrial countries, as a result of double taxation of business profits, the progressivity of tax systems and the deductability of interest, including notably mortgage interest payments (which has been found an important element depressing private saving in a number of countries, including the United States, the United Kingdom, Sweden and Denmark). In contrast, programs aiming at promoting saving through tax incentives have had little effect, and are typically distorting and costly in terms of revenue losses.

Financial liberalisation has unambiguously contributed to the poor performance of private savings in two different ways: firstly, the removal of restrictions on consumer credit in recent years has significantly increased consumer's access to credit. Secondly, financial liberalization and innovation have facilitated the process of corporate restructuring, thereby increasing leverage and capital gains by existing stockholding and dampening savings. The two described effects have not been offset by the (probably) positive effect on savings of higher interest brought about by more competition among financial intermediaries in a deregulated financial environment.

Cultural factors (like the role of bequest, and family or working patterns), quality of public institutions, and improved insurance schemes (public as well as private) have been mentioned among other possible factors that may affect saving decisions. However, the influence of each of these elements is difficult to quantify.

Table 2

PROJECTED INCREASES IN DEPENDENCY RATIOS(*) OF SEVEN INDUSTRIAL COUNTRIES

	<u>1990</u>	<u>2020</u>
Japan	48	59
Germany	44	54
Canada	47	55
united Kingdom	52	57
France	52	57
Ftaly	46	51
United States	52	55

Source: OECD projections

1)

* Proportion of people aged between 0 and 14 plus those over 65 respect to the group between 15 and 64 years old.

Research on the degree of substitutability between households and business savings clearly shows that although such substitutability exists, it is imperfect in the sense that increased saving by companies is only partly offset by a reduction in households savings. (See Musgrave and Musgrave (1984); Poterba (1987))⁷⁾. To the extent that corporate saving propensities are usually quite high, changes in taxation from corporations to individuals are one of the clearest policy tools able to promote private savings⁸⁾.

Summing up, during the 1980s net public dissaving has added to low private saving in industrial countries mainly as a consequence of financial liberalisation associated with demographic and social security developments. Available evidence points that the room for policy measures aimed at boosting private savings must be found in the areas of taxation shifts from business to individuals, the fostering of private pension funds, funded social security plans and removal of fiscal disincentives to savings. Tax incentive programs have shown to have little impact. (See Commission of the European Communities, 1990 for more references).

⁷⁾ In principle such a substitution should be perfect, since households own the business sector, whatever the form that ownership adopts. Thus, changes of wealth derived from profits or stock values should not have a different effect on private saving than that derived from other sources of income or wealth. However, the reported evidence seems to reflect a number of institutional factors, which especially affect business savings decisions.

⁸⁾ Elmeskov, Shafer and Tease (1991) point out that changing trends in business and household savings tend to offset one another over a period which widely varies among industrial countries. They suggest the possibility that it reflects "variable but offsetting measurement errors in corporate and household saving".

3. ADEQUACY OF SAVINGS: SOME THEORETICAL PERSPECTIVES AND EVIDENCE

A number of factors make it difficult to assess the adequacy of a given level of savings: firstly, the theoretical framework provided by existing growth models remains far from satisfactory. Secondly, subjective elements such as the intergenerational welfare distribution as well as the income distribution within the same generation need to be taken into account to judge the adequacy of a particular savings level. The existence of market failures and institutional distortions further complicates the issue. Finally, in an international context, an overall assessment requires that the degree of international capital mobility and institutional arrangements affecting the mobility of goods and labour be taken into account.

in the light of these caveats, any approach to judge the adequacy of savings must be cautious, and will, to a large extent, have to rely on indicators such as broad trends in saving over time, the demand for savings, the evolution of real interest rates⁹⁾, or the consequences of diverging saving behaviour among countries.

Models of economic growth provide a framework within which the long-run effects of saving may be analysed. The adequacy of a given level of saving can then be assessed in terms of the intertemporal welfare A STREET

⁹⁾ High real interest rates in the past two decades may reflect several influences, such as an improvement in the real returns of investment, uncertainty about future inflation, monetary policies aimed at controlling inflation, developments in tax systems (which combined with high inflation may involve serious distortions) or financial market deregulation (see Atkinson and Chouraqui, 1985). Barro and Sala (1990) found significant influence of stock returns and oil prices on expected real interest rates. But the influence of these factors does not preclude a major role for saving in determining the evolution of real interest rates.

distribution underlying each possible saving pattern and the trade-off involved between current and future consumption.

A fundamental test, which draws on classic growth models (based on Solow, 1956), is the well-known "golden rule", which is based on longrun consumption maximization. According to this criterion, savings must reach the level required to provide an "optimal" capital-labour ratio, in the sense that the amount of consumption which would have to be sacrificed to increase the capital stock further (via increased savings) would be greater than the increase in future consumption induced by the increase in investment. A higher level of savings (and therefore investment) would involve a reduction in current consumption greater than the increase in future consumption so attained. Thus, this rule avoids an explicit judgement on intergenerational preferences by assuming an equal weight of consumption across all generations. This criterion can be extended by including in the analysis a positive discount rate which allows for the effect of the present lower utility of delayed consumption, thus attributing a lower weight to the same amount of future consumption.

While classical growth models provide an interesting starting point, they fail to explain the key question of diverging growth rates among industrial countries. This is mainly owing to the fact that these models consider technical progress to be exogenous, despite the fact that technology generally accounts for most of the unexplained differences in growth. A related problem is that classical models imply an absence of any relationship between savings and growth in the long run, which strongly contradicts evidence for industrial countries. More recent works in this field have included technology and the growth rate as endogenous variables. Hence, according to these models, it is possible to increase the rate of growth by increasing saving and investment rates. The main difference between these and classical models lies in the role that "endogenous" models attribute to externalities associated with savings invested in areas such as human capital (Lucas, 1988), research and development, specialization of production (Romer, 1986 and 1987) and infrastructure (Barro, 1989), and with the expansion of financial intermediation as a self-feeding process which by itself boosts investment (Greenwood and Jovanovic, 1988).

The presence of a common element in these models — the existence of externalities which are not reflected in the returns obtained by private investors — involves a different (usually higher) level of socially optimal aggregate saving than that which would be generated by market forces. Therefore, those models provide theoretical support for policy measures aimed at promoting savings in so far as such measures help to attain socially optimal levels and thereby to circumvent market failures. Unfortunately, empirical evidence remains rather scarce and does not allow a precise quantification of the influence that can be attributed to each of the factors affecting future productivity. It is clear however, that, to assess the adequacy of a certain level of savings, significant attention must be payed to a number of factors which had been excluded from earlier analysis, especially human capital accumulation, research and development and infrastructure.

in addition to the above mentioned market failures, there is another broader line of arguments pointed out by Tobin (1971), which links externalities associated with capital accumulation to the achievement of low inflation, since investment involves growth, and growth in turn helps to bring about non-inflationary increases in real wages (and income redistribution). Again, these externalities to individual agents may provide good arguments to promote investment and saving beyond the levels the market itself would determine. Finally, two more important factors may cause saving rates to deviate from their optimal path: market rigidities and government policy towards saving. in particular, taxation of saving can have a negative effect (as was mentioned in section 1 above), whilst the government's influence on saving is reflected in national aggregate savings not only through the net amount itself but also through its effects on private saving ratios. since public and private savings are not perfect substitutes¹¹).

Recent empirical work has tried to evaluate the adequacy of saving ratios for individual countries in the context of the models just set out: A test of "dynamic inefficiency" is derived from the "goldenrule" concept explained above. A country will be dynamically inefficient if it can currently consume more (i.e. save less) without negatively affecting future consumption. That would mean that its current saving rate is too high. A comparison between investment and profits in a given period is a useful tool to measure dynamic inefficiency, suggested by Abel (1989). The comparison draws on the idea that the level of sacrifice (here measured by investment which in

- 16 -

¹¹⁾ According to recent estimations for industrial countries, between 25 and 40 percentage points of increases in government saving could be offset by decreases in private saving over the medium term. Hence, "Ricardian equivalence", which involves full public/private saving substitution, does not hold.

a closed economy equals saving), is rewarded by profits from the investment which must exceed that sacrifice in each period¹²). The results obtained in that study, based on the test of dynamic inefficiency, show no evidence of excess saving in any of the major industrial countries (including Japan, the country with the highest saving ratio in the past decade).

Other recent works focusing on the United States have found evidence of an insufficient saving rate. As explained above, such a conclusion requires an assumption to be made about the rate of discount attributed to future consumption. One study based on the golden-rule criterion found that the saving rate compatible with the golden rule, applying a positive rate of discount, would range between 15.5% and 30% of GDP for the period 1986-90, while the actual rate in the period has been 12.25% (Evans, 1990). A more recent study exploring the consequences of the decline in savings in the US from a historical 7.5% of GDP - in net terms - to only 3% in the 1980s evaluates alternative saving scenarios in the context of growth models: the study finds that the decline of savings in the 1980s has reduced the US capital stock by about 15% and its potential output by about 5%. If the same trends were to persist until the end of the century, the accumulated loss in capital and output would be 28% and 10%, respectively. When the same estimates are carried out in the context of a model linking capital innovation with the pace of technological innovation, losses in potential output reach 7% in 1990 and about 15% by the end of the century (Harris and Steindel, 1991).

¹²⁾ Otherwise, it would be possible by increasing current consumption (and decrease investment and saving) to attain a situation in which either sacrifice would be lower with the same future benefits (profits) or the same sacrifice but with higher profits.

Whilst the research described above provides useful references to evaluate the adequacy of the saving performance of individual countries, in the context of capital mobility the approach must be modified and new elements have to be considered. In principle, a low rate of saving in a single country may be offset by an inflow of foreign saving, provided risk-adjusted rates of return on capital are attractive enough to foreign investors. Two new questions then arise. The first, from the standpoint of the borrowing country, refers to the effects of increased external indebtedness, its sustainability and its desirability. The second question refers to the effects transmitted through the international monetary system since each country's saving/investment balance affects the overall pool of saving available and the degree of pressure on world interest rates and exchange rates.

From a theoretical perspective, a persistently low level of saving in a given country, as has been the case in the United States, may simply reflect a certain intertemporal consumption distribution, which in turn reflects individual preferences. In the context of crossborder freedom of capital movements, the excess of domestic demand for investment over national saving is reflected in a persistent inflow of capital which involves a deterioration in the country's net external asset position¹³. As far as that position is sustainable – given the willingness of foreign investors to invest in that country – it should

¹³⁾ Sinn (1990) argues that whilst theory suggests that welfare improves when a country moves from autarky to net creditor or debtor position, it does not provide a justification to use net external assets positions as targets of economic policy. He points to the case of Romania's current account surpluses, which allowed it to pay back all its foreign debt. As a consequence, "Romania's hard currency debt is no more, but there is not much left of the economy either".

not by itself¹⁴⁾ be a matter of concern. Nevertheless, from the perspective of a given country, the presence of market failures like those above described may provide good reasons to suppose that higher levels of domestic saving may be socially more desirable. From a global standpoint, as far as a persistent reliance on foreign saving may crowd out investment in other countries and put pressure on prevailing world interest rates, such concern may be justified, particularly when the saving shortfall occurs in highly developed countries.

This conclusion is supported by scenarios generated with the MULTIMOD model of the IMF (see IMF, 1991b), which show the potential positive effects of increased public saving in industrial countries. Such positive effects are reflected both in generalized medium term interest rate reductions, and increases in output (see table 3).

Even if the assumptions on which these projections are based might not be met (in terms of fiscal consolidation), their results highlight the widespread transmission of the saving behaviour of each country across the world, and the potential global benefits which could be obtained from a return to higher rates of saving of the industrial countries. Section 6 below discusses further the cross-border effects of national savings and its recent patterns.

¹⁴⁾ The results of the studies by Feldstein and Horioka (1980) and Feldstein and Bachetta (1989) suggest that an increase in national saving does raise the nation's capital stock, even in a context of relative openess of capital flows.

\sim	
-	
~	
- cr	
~~~	
-	
æ	
-	
5	
-2	
0	
ు	
- 1	
~	
~	
-	
æ	
h	
ŝ	
~	
~	
~	
-	
~	
_	
6	
ž	
-	
2	
_	
ഗ	
0	
~~~	
လွ	
~	
تيا	
9	
~	
~	
-	
4	
3	
-	
10	
်	
-	
്	
يت ا	
ч	
-	
œ	
111	
-	
-	
-	
- 22	
\rightarrow	
-	
-	
_	
- 22	
Ē	

(Percentage points variations respect to IMF World Economic Outlook baseline projections, period 1991-96)

	Nationa	(Saving #	odification)	Effects on						
	Total	Private	Government	Foreign saving	Investment	Real Business Fixed Investment	Interes Short-ter m	t Rates Long-term	Capital stock	Potential Output
United States	1.3	-0.7	1.2	-0.5	0.8	0.7	-1.2	-1.2	1.9	0.6
Italy	1.2	-0.8	2.0	-0.3	0.9	0.8	-0.9	-1.0	1.7	0.6
Canada	0.8	-0.4	1.2	ı	0.8	0.7	-1.4	-1.3	1.7	0.6
Germany (West)	0.3	-0.1	0.4	0.5	0.8	0.7	-1.0	-1.0	1.5	0.6
<u>All Industrial</u> countries	0.9	-0.2	1.1	ı	0.8	0.6	-1.1	-1.1	1.5	0.5

(1) Projections based on the IMF's MULTIMOD model, assuming that measures of fiscal consolidation projected by governments are met over the period, In particular for the countries especified in the table.

Source: International Monetary Fund (1991)

4. ORIGIN VERSUS ALLOCATION OF SAVINGS

In a context of increasing financial integration, any assessment of the adequacy of a certain level of savings inevitably involves a judgement about the appropriateness of the international allocation of world savings. Some important trade offs are involved in this regard. On the one hand, integration of financial markets permits an enhancement in the allocation of resources accross countries, allowing market forces to determine in which country other countries savings are to be invested. On the other hand, with an integration of financial markets policy makers face fewer constraints in the short run in policy decisions that affect each country's contribution to world savings; they may therefore be tempted to tap into the pool of world savings to finance projects with a very low social rate of return.

The removal of capital controls experienced in recent years among most industrial countries has been a major factor boosting international financial integration. However, several other circumstances besides capital controls determine the degree of financial integration. Among them, exchange rate developments are the single most important factor. In addition, institutional arrangements (such as the liberalization of legislation concerning the cross-border establishment of financial institutions, and the harmonization of prudential and regulatory arrangements) and the presence of market failures (basically owing to information shortages), play a role.

A possible approach to the measure of integration of capital markets is the size of gross and net capital flows. Both in net terms (as shown in recent years by major current account imbalances), and especially in gross terms, the increase in capital flows experienced in recent years has been considerable. Between 1979 and the period 1986-1988, annual capital outflows from the Group of Seven countries expanded by some 250% nominally (in SDR terms), which implies a real increase of around 200% (Latter and Allen, 1990). In gross terms, indicators such as cross-border banking expansion (see chart 3), international transactions in securities, or the development of Euromarkets, all show huge rates of expansion.

In addition to the scale of capital flows, two more approaches have been used to assess the degree of financial integration in recent years¹⁵): one based on risk-adjusted interest rate differentials, and another based on the degree of correlation found between saving and investment ratios of different countries¹⁶).

Evidence of a tendency for interest rate differentials to narrow seems to confirm that the degree of financial integration has increased considerably since the mid-1970s. Using this approach, Frankei (1989) concluded that by the end of the 1980s financial markets were virtually completely integrated among the large industrial countries. More recent studies generally support that view, even if it appears more evident for short-term than for long-term capital flows.

¹⁵⁾ it must be noted that the size of capital flows may not be a good indicator of financial integration. Highly integrated financial markets may be reflected in close movements of asset prices but small capital flows. Conversely, relatively large capital flows do not necessarily imply a high degree of financial integration, but may reflect the evolution of fundamental factors (differences in productivity), which would warrant high capital flows.

¹⁶⁾ See Goldstein et.al. (1991) for a review of evidence on integration of major financial markets.

```
GRAPH 3 - EXPANSION OF INTERNATIONAL
FINANCIAL ACTIVITY
(International bank lending, US $ bill.)
```



Source: B.I.S.

Alternative approaches to judge the degree of financial integration include the evolution of interest rate differentials for the same currency between domestic and offshore markets, and developments in parity of interest rate differentials, either covered or uncovered¹⁷⁾.

With regard to the first approach, a decreasing trend in differentials between domestic and offshore markets has been evident in recent years, with marked reductions in countries gradually removing capital controls. The approach using covered interest rate parity confirms the impression of a high degree of integration among industrial countries, especially for short-term instruments (see Frankel, 1991).

Finally, analysis of saving-investment correlations in individual countries, has generally shown a higher degree of correlation than expected, which might suggest a smaller degree of capital market integration than indicated by other types of research¹⁸⁾. Early findings by Feldstein and Horloka (1980) pointed to a fairly high degree of correlation between saving and investment in the industrial countries in the 1970s. Further work by Feldstein and Bachetta (1989) confirmed those results, but nevertheless revealed an increase in international capital mobility along the 1980s. This research was based on an examination of the savings retention coefficient, which was 0.79 in the period 1980-86, down from 0.91 in the 1960s and 0.86 in the 1970s.

¹⁷⁾ Covered interest rate differentials are provided by the difference between interest rates on instruments issued by comparable borrowers denominated in different currencies, after adjusting those differences for the cost of cover in forward exchange markets.

¹⁸⁾ This seems to hold for both industrial and developing countries. See Dooley, Frankel and Mathleson (1987).

Other studies have attempted to explain the high saving/investment correlations by examining private sector behaviour (i.e. trends in productivity, or demographic factors which simultaneously affect saving and investment). Perhaps the most plausible explanation is that the high correlation seems to reflect the impact of government policies aimed at reducing current account imbalances¹⁹).

As mentioned earlier, capital mobility is a result of several factors besides capital controls. Financial deregulation has been a rather recent phenomenon in major industrial countries, and high exchange rate volatility continues to appear as a major factor hampering capital mobility, especially for long-term flows²⁰.

Summing up, there are a number of indications that capital mobility has considerably increased in industrial countries during the 1980s, in spite of the persistence of some constraining factors, and the trend is likely to continue in the years ahead. A major factor that continues to dampen the degree of mobility will be the exchange rate volatility. Since relatively large shares of countries' savings continue to be retained domestically, there would seem to be ample scope for each country to influence the level of savings available for domestic investment through its domestic economic policies.

¹⁹⁾ See Artis and Bayoumi (1989). They found strong negative contemporanean correlations between the saving-investment balances of the government and private sectors in the 1970s and somewhat lower correlations in the 1980s, which are nevertheless compatible with relatively low capital mobility.

²⁰⁾ This seems to be confirmed by the lower saving/investment correlation - and hence a higher degree of financial integration that is apparent among the countries participating in the exchange rate mechanism of the European Monetary System, as explained in section 6 below.

5. CURRENT ACCOUNT IMBALANCES AND CAPITAL FLOWS

The progressive removal of capital controls and international financial deregulation experienced in the 1980s should in principle be welcomed, since increased capital mobility is a key factor to achieve a better international allocation of resources. Under conditions of a high degree of capital mobility, one can expect the market to determine the sustainability of current account imbalances either by providing financing to a deficit country, or by reducing the flow of capital, thereby creating the incentives for domestic policies to adjust the imbalance.

As far as current account imbalances merely reflect demographic factors, or more generally, intertemporal consumption optimization decisions within each country, the consequence of free capital mobility is that private agents in the rest of the world -- be it direct investors or financial intermediaries -- perform their role of assessing alternative investment opportunities at the global scale, with the consequent gains in efficiency.

These arguments are widely recognised as advantages from increased capital mobility, but some caveats nevertheless must be made with regard to the potential dangers involved. As far as institutional barriers and market failures persist --- as is still the case to a large extent even among most integrated industrial countries --- advantages from capital mobility may not be fully realized, and some negative consequences might even appear. From a domestic perspective, a current account deficit may not be a problem by itself if it can be financed through private capital inflows. Nevertheless, as far as that deficit reflects either market failures, regulating distortions or an excessive government deficit (when compared with a socially optimal level), external financing may serve to delay fundamental adjustments, thereby increasing the eventual costs of subsequent adjustment policies²¹). If this was the case, the negative effects could be transmitted to other countries, particularly in a context of a high degree of economic interdependence and financial integration²²).

Relationship between the current account and capital flows

While significant current account imbalances have arisen in other periods of the past, there are a number of key features which distinguish them and the involved capital flows pattern from those experienced in the second half of the 80s.

During the gold standard (XIX century until 1914) significant current account imbalances followed the pattern of capital outflows from rich and slow-growing countries (basically England) to relatively poor and fast growing countries (notably the United States). In a context of secular low inflation, issuance of fixed interest long-term bonds was

²¹⁾ A good example of markets' wrong perception of the sustainability of a lending pattern and its potential negative consequences is provided by the debt crisis. The adjustment costs, especially for heavily borrowing countries, are likely to last for a number of years ahead.

²²⁾ These arguments have implications for the appropriate order in which capital and factors' transactions should be liberalized. Frenkel (1982) emphasised the dangers involved in liberalizing liberalization of capital transactions without а previous commercial transactions. Such a sequence of liberalization involves a higher risk of distortions in resource allocation, and a higher cost of such potential distortions owing to the inadequate adjustment of fundamentals, and to the fact that capital flows (potentially much larger than commercial transactions) are likely to overreact to "wrong" price signals.

the usual channel for those flows. Thus, the pattern of capital flows was a natural one, as far as saving surpluses were channelled into investment in countries with a higher rate of return on capital, which provided a sound and stable source to service the country's debt

The 1970s marked a more recent period of large current account imbalances as a consequence of the oli crisis. Euring that period. Imbalances emerged rather homogeneously in virtually all oli importing countries. The pattern followed by capital flows was shaped by the intensity of the real shocks which gave rise to the imbalances, with huge outflows of capital from oll exporting to the major financial centers. Through "recycling" these capital flows were subsequently channelled through the international banking system and the emerging euromarket to deficit countries in both the industrial and developing world. However, it is now widely recognised that the scale of these capital flows did not always take account of economic fundamentals, as they often financed consumption or investment projects that proved not to be viable. These capital flows therefore were a major cause behind the subsequent debt crisis.

During the 1980s, and especially from 1982, financial liberalisation and removal of capital controls in conjunction with the emergence of big budget deficits in some industrial countries (notably the United States), again gave rise to very large current account imbalances among the major industrial countries (see Chart 5).

in addition to the emergence of unprecedented (since the gold standard) current account imbalances among the major industrial countries, the



GRAPH 4a - CURRENT ACCOUNT IMBALANCES OF MAJOR INDUSTRIAL COUNTRIES (% of GDP)

GRAPH 4b - CURRENT ACCOUNT IMBALANCES OF MAJOR INDUSTRIAL COUNTRIES (% of GDP)





GRAPH 5 - CURRENT ACCOUNT IMBALANCES OF MAJOR INDUSTRIAL COUNTRIES (*)

(*) Weighted average of the sum of deficits and surpluses of G-7 countries, as a % of GDP).

dramatic increase in the scale of international capital movements has been accompanied by an increasing loss of links between those movements and trends observed in trade or other economic fundamentals. These developments have been linked to the increasing sophistication of financial markets and the sharp decline in information and transaction costs that has occurred, creating a truly global financial market.

The dynamic so generated has made it more difficult to identify the rather unclear causality between current account imbalances and capital flows. However, there has clearly been a movement in the direction of a greater simultaneity in the determination of both. In contrast to the gold standard period, however, capital flows are not seen merely as the counterpart to profitable investment opportunities abroad²³). In fact, there seems to have been a shift in the pattern of capital inflows towards countries with low savings but also with relatively low rates of growth and returns on capital, thereby modifying the "natural" pattern. Moreover, there is evidence that capital flows may have contributed to the widening of current account imbalances, and not merely to their financing.

Nevertheless, the difficulty in identifying the factors that determine both net and gross capital flows has been widely recognized, since the reaction of markets to shocks can be reflected in capital flows, in

²³⁾ A related discussion is the adoption of one of the two possible approaches to the determination of current account balances: the "alasticities" approach (based on relative prices) and the "absorption" apporach (based on saving-investment imbalances). Since both processes take place simultaneously, the two approaches are compatible, and essentially describe the same process.

asset price adjustments, or in a combination of the two²⁴⁾. However, it is now clear that, besides the above mentioned loss of links between capital flows and developments of the real economy, a number of financial factors can be identified behind the sharp increase experienced by capital flows. Among them, the degree of financial sophistication reached both in domestic and international markets, the removal of barriers to financial integration, and the increased level of competition among financial intermediaries should be mentioned.

In addition to the factors already mentioned, the increasing needs of private agents to hedge against financial risks (be it exchange rate risk or interest rate risk), and the increased possibilities open to arbitrage and speculative operations have undoubtedly contributed to the rise in gross capital flows. Besides this, other explanations can be found behind the large increase observed in international portfolio and direct investment. On the one hand, hedging against risks involved in investment concentration has led in recent years to a process of portfolio diversification which is not yet completed²⁵). On the other hand, commercial strategies and the need to circumvent barriers (either trade, fiscal or financial) have also contributed to boost direct investment in recent years.

²⁴⁾ Nevertheless, some fundamental and regulatory factors can easily be identified as playing an important role in explaining large capital flows during the past two decades, such as big capital outflows from Germany in 1987-89, owed to the planned introduction of a witholding tax on interest income, or the large capital inflow in the US during 1981-85, related to sharp differences in fiscal/monetary policy mix respect to Japan and Germany, among other examples. (See IMF, 1991).

²⁵⁾ Turner (1991) argues that in the 1990s this process should slow its pace, given the virtual completion of the process of financial liberalisation and the associated diversification of assets.

In summary, there are a number of arguments which suggest that the liberalization of capital movements and the sharp increase in international capital flows may not have produced all the potential benefits suggested by economic theory. The most negative aspects of recent changes in the patterns of international allocation of savings has been the extent to which it has allowed to postpone fundamental policy adjustment through increased access to international capital markets.

6. SAVING IN THE EUROPEAN COMMUNITY IN THE CONTEXT OF ECONOMIC AND MONETARY UNION

Efforts to stimulate savings and enhance its allocation across countries are at the heart of the process towards Economic and Monetary Union. The convergence required in the field of fiscal consolidation, as a pre-condition for the final stage of EMU, constitutes a clear incentive to promote public sector saving. In addition, the completion of the Single Market involves a progressive liberalization of capital movements and provides a framework for a market-based allocation of saving throughout the Community.

There is abundant evidence of the impact of the European Monetary System on capital market integration. Feldstein and Bachetta (1989) found comparatively low saving/investment correlations for EMS countries suggesting a high degree of integration. Results obtained by Bandhari and Mayer (1990) have confirmed that there is a higher degree of capital mobility among EMS countries than among other industrial countries²⁶).

Data on the composition of capital flows in the Community suggest that there has recently been a significant increase in foreign direct investment, which in 1989 exceeded \$7 billion, exceeding for the first time in recent years the amount of direct investment in the United States (see Turner, 1991). This development has been linked to a considerable increase in intra-EC mergers and acquisitions (see Commission of the European Communities, 1989). These trends indicate the intensity of the process of financial adjustment that has been generated by the Single Market and EMU processes. The conditions have thereby improved for significant gains in the efficiency with which savings are allocated across the Community.

At the same time, the discussion in this paper has highlighted some of the potential dangers that may result from the liberalization of capital flows within the Community. In this context, it is obvious that the success of the EMU process to a large extent will depend on the degree of "discipline" accepted by each member country in terms of saving. Developments in budget deficits will be of paramount importance, which explains the emphasis on this issue in the context of the convergence process. The danger, of course, is that monetary

²⁶⁾ A recent comparison of capital mobility in different EC countries has found significant differences in capital mobility for the period 1960-88. Germany, UK and the Netherlands are the countries with the highest degree of capital mobility (Argimón and Roldán, 1991). Since capital controls have been the main factor explaining those differences, they should be reduced in the coming years.

integration in the absence of firm rules for fiscal policy will strengthen the ability for individual countries to finance their budget deficits, and thereby encourage fiscal laxness.

More generally, given the Community's potential role as a capital exporter (to Eastern Europe, the Soviet Union and the development world) an appropriate functioning of the process should be reflected in the achievement of a net saving surplus for the Community, in a context in which each EC country provides an "adequate" contribution to the Community's pool of savings. In this respect, it might be appropriate to think in terms of some kind of "saving targeting", as the EMU process gradually reduces the current account constraint.

7. CONCLUSIONS

Concerns about low saving levels in industrial countries during the past decade seem to be justified, particularly in the presence of a foreseeable increase in investment demand during the 1990s. Although there are few signs that the low level of saving has had major repercussions on economic performance so far, the inadequacy of savings risks becoming a decisive constraint on growth in the future, particularly for the newly liberalizing countries in Central and Eastern Europe and the developing world.

in the current context of relatively high (and increasing) financial integration, the negative effects of low saving in individual countries are transmitted to the rest of the world. This constitutes a systemic deficiency to the extent that the pattern of capital flows prevailing in recent years has tended to offset the impact of low saving propensities rather than promote policy changes. Capital market integration may therefore delay required adjustments. An increased international consensus to promote private and public saving may thus be necessary to achieve potential advantages of capital market integration.

Additional negative effects stemming from capital market liberalization include a shift from investment to consumption financing and the associated negative consequences in terms of crowding out of profitable investment opportunities through higher interest rates. In this respect, the existence of market failures (externalities) affecting investment suggests that the market by itself does not guarantee adequate levels of saving and that public intervention in this field consequently may be warranted.

Among the negative systemic consequences of low saving and high interest rates, the effects on developing countries are particularly serious. With inadequate levels of domestic saving in the developing countries themselves and growing difficulties in attracting funds from abroad, the growth performance of many developing countries during the past decade has been extremely weak. Barring a substantial increase in world saving, there is little prospect of any recovery of growth in many of these countries over the medium term.

The low level of saving also constitutes a constraint on the Western world's ability to assist Eastern Europe and the Soviet Union in their reform efforts. While a high level of domestic saving in these countries and greater efficiency in the allocation of their saving are

key conditions for the restructuring efforts to succeed, a certain transfer of resources from the rest of world will also be necessary to complement domestic savings. With the low saving levels in the West, however, there is a risk that the transfer of resources to Eastern Europe and the Soviet Union will increase pressures on interest rates worldwide and crowd out investment in other countries, notably the developing countries. Such a situation would also increase pressures to expand officially guaranteed (and thereby subsidised) credits, which would further increase interest rates in private markets.

Summing up, as high financial integration has reduced the significance of the current account as an economic target or constraint, international policy coordination will have to provide for an appropriate degree of discipline by ensuring that incentives to saving are adequate. In addition to measures aimed at promoting private savings (see page 14), some kind of public sector saving targeting will be necessary to ensure an adequate level of saving worldwide.

Within the Community, some of the advantages of Economic and Monetary Union in terms of increased capital mobility have already been achieved, mainly owing to the reduction of exchange rate uncertainty, which has been identified as a major factor constraining capital mobility outside the European Monetary System. Nevertheless, to fully reap the potential benefits from Economic and Monetary Union "rules" will have to be implemented to ensure both an adequate level of savings overail in the Community and an appropriate degree of convergence of saving propensities across Community member countries.

ANNEX

STATISTICAL ISSUES ON SAVING AND CURRENT ACCOUNT

When analysing conventional figures of saving, two kinds of caveats must be made: firstly, national accounting criteria are not homogeneous among countries. Secondly, even when analysing data of a given country the assumptions underlying conventional data must be taken into account before drawing any conclusions. Diverging accounting criteria can explain a part of the large differences in absolute levels of saving observed among industrial countries, even if the broad trends in saving described in section 2 above do not seem to significantly change when most usual adjustments are carried out. This annex briefly enumerates the main adjustments which have been suggested for conventional measures of saving. Finally, the issue of the well known statistical current account discrepancy is addressed, which during the last decade has raised doubts on the adequancy of the available data.

Measuring saving

According to the United Nations' System of National Accounts (SNA), saving is calculated for each of the institutional sectors (i.e., households, enterprises and government) as a residual, by substracting current outlays from income. The United States and some European countries have not adopted the United Nations system. While in all systems current income and expenditure are measured in a homogeneous way, a major difference in the United States' system is the fact that all government disbursement is treated as government consumption, while the SNA system considers government expenditure in construction and non-military equipment as capital expenditure, and therefore the resulting measured saving is considerably lower than what would be according to SNA standards (see Aghevil et.al (1990) and Elmeskov, Shafer and Tease (1991) for a review of these issues and a summary of the effects of possible adjustments).

In addition to the former, the relative size of the underground economy, and a number of conceptual problems can significantly modify the conclusions obtained when analysing conventional data. Among those problems, the treatment of consumer durables, of capital gains and losses¹⁾, depreciation and the distinction between real and nominal interest payments have deserved major attention. As pointed by Smith (1989) : "When intercountry comparisons are made, it is important to identify the purpose of the comparison and determine whether the measure being used is suited to that purpose".

Elmeskov, Shafer and Tease (1991) classify possible adjustments to conventional measures of saving as follows: changes in asset values (including the effects of inflation and capital gains); changes in the classification of economic activities (which affects the definitions of income and consumption); and the inclusion of activities not covered by national accounts but which affect future income and welfare. Adjustments provided in this work show to what extent the variations with respect to conventional data can be significant. For instance, inflation adjusted private saving ratios show differences of 4.8 and 4.9 percentage points of variation between the 1970s and the 1980s for the cases of the United Kingdom and Japan. Likewise, if expenditure in education or in research and development are considered as "saving" the results obtained vary significantly in most developed countries. The conclusions of their work point that while most adjustments to gross saving (such as the mentioned inclusion of Research and Development or education disbursement) would boost national saving ratios, other adjustments like the deduction of depreciation and of the depletion of resources would result in much lower net saving ratios than those measured by SNA criteria. Nevertheless, they conclude that in spite of the uncertainties and errors involved on measurement issues, "the general picture of the 1980s as a period of weak saving in the OECD area, at least until the last years of the decade, seems relatively robust".

The current account discrepancy

The fact that the world's current account data show a large aggregate net debt since the early 1980s has been a matter of concern and the subject of several ad-hoc studies aimed at determining the origins of such discrepancies and the extent to which greater accounting

¹⁾ See Harris and Steindel (1991) for a discussion on the definition of productive investment and the adequacy of stock market as a measure of saving.

homogeneity among different countries could help to overcome that problem. The major source of statistical errors which has been identified behind the global current account discrepancy is the emergence of large amounts of cross-border assets recognised by debtor countries but not by creditors. Other indentified sources of discrepancies include the shipping and unrequited transfer sectors of the current account balance, as well as problems of geographic allocation.

While the persistence of a significant global discrepancy remains a source of concern and caveats when analysing trends in global flows, the fact that developed countries became net capital importers before that discrepancy appeared suggests that even if the magnitude of real net flows is unknown, its actual direction is that shown by available statistics. Furthermore, it has been found that most of the identified adjustments which would correct the global discrepancy would not be concentrated enough in any single country or area as to put under question the basic conclusions obtained from uncorrected data²).

<u>Table : The world's current account discrepancy</u> (As a percentage of total exports of goods and services)

<u>1983</u>	<u>1984</u>	<u>1985</u>	<u>1986</u>	<u>1987</u>	<u>1988</u>	<u>1990</u>
-2.4	-2.4	-3.0	-2.0	-1.2	-1.6	-2.3

Source : World Economic Outlook, IMF May 1991.

²⁾ See IMF (1987) and IMF (1989) for a detailed analysis of the statistical problems.

REFERENCES

- Abel, Andrew B., N. Gregory Mankiw, Lawrence H. Summers, and Richard J. Zeckhauser, "Assessing Dynamic Efficiency : Theory and Evidence," Review of Economic Studies, Vol. 56 (January 1989), pp. 1-19
- Aghevil, Bijan B., James M. Boughton, Peter J. Montiel, Dellano Villnueva and Geoffrey Woglom, "The role of national saving in the world economy. Recent trends and prospects". IMF Occasional Paper n° 67 (Washington, International Monetary Fund, March 1990)
- Ando, Albert, and Franco Modigliani, "The 'Life Cycle' Hypothesis of Saving: Aggregate Implications and Tests," American Economic Review, Vol.53 (March 1963), pp. 55-84
- Argimón, Isabel, and Jose María Roldán, "Ahorro, Inversión y movilidad internacional del capital en los países de la CE". Boletin Economico del Banco de España (Madrid, Banco de España, April 1991) pp. 39-43
- Artis, Michael, and Tamim Bayoumi, "Saving, Investment, Financial integration, and the Balance of Payments," IMF Working Paper, n° 89/102 (Washington: International Monetary Fund, December 14, 1989)
- Atkinson, Paul, and Jean-Claude Chouraqui, "The origins of high real interest rates" OECD Economic Studies n° 5 (Paris : Organisation for Economic Cooperation and Development, 1985)
- Bandhari, Jadgeep S. and Thomas S. Mayer. "A note on saving-investment correlation in the EMS. IMF Working Paper n° 90/97 (Washington: international Monetary Fund, October 1990)
- Barro, Robert J. "The neoclassical approach to fiscal policy", in Modern business cycle theory, ed by Robert J. Barro (Chicago, 1989) pp. 178-235
- Barro, Robert J. and Xavier Sala i Martin, "World real interest rates" NBER Working Paper n° 3317 (Cambridge, Massachusetts : National Bureau of Economic Research, 1990)
- Commission of the European Communities, "Saving, investment and real interest rates". Study n°8, European Economy n°46, (Brussels, Commission of the European Communities, December 1990)
- Commission of the European Communities, "The effects of the internal market". Study n⁵, European Economy n⁴², (Brussels, Commission of the European Communities, November 1989)
- Dean, Andrew, Martine Durand, John Fallon and Peter Hoeller, "Saving trends and behaviour in OECD countries". OECD Economic Studies n° 14 (Paris : Organisation for Economic Cooperation and Development, Spring 1990), pp. 7-88
- Dooley, Michael, Jeffrey Frankel, and Donald J. Mathieson. "International capital mobility : what do saving-investment correlations tell us?" IMF Staff Papers, Vol. 34, Washington, International Monetary Fund, (September, 1987)

- Elmeskov, Jorgen, Jeffrey Shafer and Warren Tease, "Savings trends and measurement issues", OECD Working Paper n° 105 (Paris: Organisation for Economic Cooperation and Development, 1991)
- Evans, Owen. "National Savings and targets for the Federal Budget balance in the United States". IMF Working Papers 89/103 (Washington : International Monetary Fund, December 1989)
- Feidstein, Martin S., and Charles Horloka, "Domestic Saving and International Capital Flows," Economic Journal, Vol. 90 (June 1980), pp. 314-29
- Feldstein, Martin S., and Philippe Bacchetta, "National Saving and International Investment," Discussion Paper n° 1463 (Cambridge, Massachusetts: Harvard Institute of Economic Research, November 1989)
- Frankel, Jeffrey A. "Quantifying international capital mobility in the 1980s" in, National saving and economic performance, ed. by Douglas Bernheim and John B. Shoven (Chicago, University of Chicago Press, June 1991)
- Frankel, Jeffrey A. "Quantifying international capital mobility in the 1980s," NBER Working Paper, n° 2856 (Cambridge, Massachusetts : National Bureau of Economic Research, 1989)
- Frenkel, Jacob. "The Order of Economic liberalisation : discussion" in Economic policy in a world of change, ed by K. Brunner and A.H. Meitzer Carnegie, Rocester conference series on public policy (North Holland 1982)
- Goldstein, Morris, Donald J. Mathieson and Timothy Lane, "Determinants and Systemic Consequences of international capital flows", in Determinants and Systemic consequences of international capital flows, IMF Occasional Paper n° 77 (Washington : International Monetary Fund, March, 1991)
- Graham, John W. "International differences in saving rates and the life cycle hypothesis". European Economic Review, Vol 31 (1987) pp. 1509-1529
- Green, Francis, "Institutional and other unconventional theories of saving". Journal of Economic Issues. Vol XXV n° 1 (March 1991) pp. 93-113
- Greenwood, Jeremy, and Boyan Jovanovic, "Financial Development, Growth, and the Distribution of Income," Working Paper n° 131, Rochester Center for Economic Research (May 1988)
- Harris, Ethan S., and Charles Steindel, "The decline in US saving and its implications for economic growth". FRBNY Quarterly Review (New York: Federal Reserve Bank of New York, Winter 1991) pp 1-19
- International Monetary Fund, World Economic Outlook : a survey by the staff of the International Monetary Fund (Washington : International Monetary Fund, May 1991)

- Kauffmann, Barbara, "Saving's behaviour of private households in the United States and West Germany". Jahrbücher für Nationalökonomie und Statistik, vol 207, (Stuttgart, Gustav Frischer Verlag, 1990) pp 97-108
- Latter, Anthony and Heilen Allen. "International private capital flows, current account Imbalances and exchange rates", in "International capital flows, exchange rate determination and persistent current account Imbalances" BIS (Basie : Bank for International Settlements, 1990) pp 255-284
- Lucas, Robert E., Jr., "On the Mechanisms of Economic Development", Journal of Monetary Economics, Vol 22 (July 1988), pp 3-42
- Musgrave, Richard A. and Peggy B. Musgrave, Public Finance in theory and practice (new York, Mc Graw Hill, 1984)
- Poterba, James M., "Tax policy and corporate saving", Brookings Papers on Economic Activity 2, (1987), pp 455-503
- Romer, Paul M., "Increasing Returns and Long-Run Growth," Journal of Political Economy, Vol. 94 (October 1986), pp.1002-37
- Romer, Paul M., (1987), "Growth Based on Increasing Returns Due to Specialisation," American Economic Review, Papers and Proceedings of the Ninety-Nonth Annual Meeting of the American Economic Association Vol. 77 (May 1987), pp 56-62
- Sinn, Stefan, "Net external asset position of 145 countries. Estimation and interpretation", Voi 234 Kieler Studien (Tübingen : Institut für Weltwirtschaft an der Universität Kiel, 1990)
- Smith, Roger S., "Factors Affecting Saving, Policy Tools, and Tax Reform: A review," IMF Working Paper, n° 89/47 (Washington : International Monetary Fund, May 23, 1989)
- Solow, Robert M., "A Contribution to the Theory of Economic Growth," Quarterly Journal of Economics, Vol 70 (February 1956), pp 65-94
- Tesar, Linda L., "Savings, investment and international Capital Flows," Brookings Discusion Papers in International Economics, n° 64 (Washington: The Brookings Institution, June 1988)
- Tobin, James, "Economic Growth as an Objective of Government Policy," in Essays in Economics : Macroeconomics, Vol. 1 (New York : North Holland, 1971)
- Turner, Philip, "Capital flows in the 1980s : a survey of major trends". BIS Economic Papers n' 30 (Basie, Bank for International Settlements, Monetary and Economic Department, April 1991)
- Wilcox, David W, "Household spending and saving : measurement, trends and analysis". Federal Reserve Bulletin (Washington : Board of Governors of the Federal Reserve System, January 1991) pp 1-17

Economic Papers

The following papers have been issued. Copies may be obtained by applying to the address mentioned on the inside front cover.

- No. 1 EEC-DG II inflationary expectations. Survey based inflationary expectations for the EEC countries, by F. Papadia and V. Basano (May 1981).
- No. 3 A review of the informal economy in the European Community, by Adrian Smith (July 1981).
- No. 4 Problems of interdependence in a multipolar world, by Tommaso Padoa-Schioppa (August 1981).
- No. 5 European Dimensions in the Adjustment Problems, by Michael Emerson (August 1981).
- No. 6 The bilateral trade linkages of the Eurolink Model : An analysis of foreign trade and competitiveness, by P. Ranuzzi (January 1982).
- No. 7 United Kingdom, Medium term economic trends and problems, by D. Adams, S. Gillespie, M. Green and H. Wortmann (February 1982).
- No. 8 Où en est la théorie macroéconomique, par E. Malinvaud (juin 1982).
- No. 9 Marginal Employment Subsidies : An Effective Policy to Generate Employment, by Carl Chiarella and Alfred Steinherr (November 1982).
- No. 10 The Great Depression : A Repeat in the 1980s ?, by Alfred Steinherr (November 1982).
- No. 11 Evolution et problèmes structurels de l'économie néerlandaise, par D.C. Breedveld, C. Depoortere, A. Finetti, Dr. J.M.G. Pieters et C. Vanbelle (mars 1983).
- No. 12 Macroeconomic prospects and policies for the European Community, by Giorgio Basevi, Olivier Blanchard, Willem Buiter, Rudiger Dornbusch, and Richard Layard (April 1983).
- No. 13 The supply of output equations in the EC-countries and the use of the survey-based inflationary expectations, by Paul De Grauwe and Mustapha Nabli (May 1983).
- No. 14 Structural trends of financial systems and capital accumulation : France, Germany, Italy, by G. Nardozzi (May 1983).
- No. 15 Monetary assets and inflation induced distorsions of the national accounts conceptual issues and correction of sectoral income flows in 5 EEC countries, by Alex Cukierman and Jorgen Mortensen (May 1983).
- No. 16 Federal Republic of Cormany. Medium-term economic trends and problems, by F. Allgayer, S. Gillespie, M. Green and H. Wortmann (June 1983).
- No. 17 The employment miracle in the US and stagnation employment in the EC, by M. Wegner (July 1983).
- No. 18 Productive Performance in West German Manufacturing Industry 1970-1980; A Farrell Frontier Characterisation, by D. Todd (August 1983).
- No. 19 Central-Bank Policy and the Financing of Government Budget Deficits : A Cross-Country Comparison, by G. Demopoulos, G. Katsimbris and S. Miller (September 1983).
- No. 20 Monetary assets and inflation induced distortions of the national accounts. The case of Belgium, by Ken Lennan (October 1983).
- No. 21 Actifs financiers et distorsions des flux sectoriels dues à l'inflation : le cas de la France, par J.-P. Baché (octobre 1983).
- No. 22 Approche pragmatique pour une politique de plein emploi : les subventions à la création d'emplois, par A. Steinherr et B. Van Haeperen (octobre 1983).
- No. 23 Income Distribution and Employment in the European Communities 1960-1982, by A. Steinherr (December 1983).
- No. 24 U.S. Deficits, the dollar and Europe, by O. Blanchard and R. Dornbusch (December 1983).
- No. 25 Monetary Assets and inflation induced distortions of the national accounts. The case of the Federal Republic of Germany, by H. Wittelsberger (January 1984).

- No. 26 Actifs financiers et distorsions des flux sectoriels dues à l'inflation : le cas de l'Italie, par A. Reati (janvier 1984).
- No. 27 Evolution et problemes structurels de l'économie italienne, par Q. Ciardelli, F. Colasanti et X. Lannes (janvier 1984).
- Nc 28 International Co-operation in Macro-economic Policies, by J.E. Meade (February 1984).
- No 29 The Growth of Public Expenditure in the EEC Countries 1960-1981 : Some Reflections, by Douglas Todd (December 1983).
- No. 30 The integration of EEC qualitative consumer survey results in econometric modelling : an application to the consumption function, by Peter Praet (February 1984).
- No. 31 Report of the CEPS Macroeconomic Policy Group. EUROPE : The case for unsustainable growth, by R. Layard, G. Basevi, O. Blanchard, W. Buiter and R. Dornbusch (April 1984).
- No. 32 Total Factor Productivity Growth and the Productivity Slowdown in the West German Industrial Sector. 1970-1981, by Douglas Todd (April 1984).
- No. 33 An analytical Formulation and Evaluation of the Existing Structure of Legal Reserve Requirements of the Greek Economy : An Uncommon Case, by G. Demopoulos (June 1984).
- No. 34 Factor Productivity Growth in Four EEC Countries, 1960-1981, by Douglas Todd (October 1984).
- No. 35 Rate of profit, business cycles and capital accumulation in U.K. industry, 1959-1981, by Angelo Reati (November 1984).
- No. 36 Report of the CEPS Macroeconomic Policy Group. Employment and Growth in Europe : A Two-Handed Approach by P. Blanchard, R. Dornbusch, J. Drèze, H. Giersch, R. Layard and M. Monti (June 1985).
- No. 37 Schemas for the construction of an "auxiliary econometric model" for the social security system, by A. Coppini and G. Laina (June 1985).
- No. 38 Seasonal and Cyclical Variations in Relationship among Expectations, Plans and Realizations in Business Test Surveys, by H. König and M. Nerlove (July 1985).
- No. 39 Analysis of the stabilisation mechanisms of macroeconomic models : a comparison of the Eurolink models by A. Bucher and V. Rossi (July 1985).
- No. 40 Rate of profit, business cycles and capital accumulation in West German industry, 1960-1981, by A. Reati (July 1985).
- No. 41 Inflation induced redistributions via monetary assets in five European countries : 1974-1982, by A. Cukierman, K. Lennan and F. Papadia (September 1985).
- No. 42 Work Sharing : Why ? How ? How not ..., by Jacques H. Drèze (December 1985).
- No. 43 Toward Understanding Major Fluctuations of the Dollar by P. Armington (January 1986).
- No. 44 Predictive value of firms' manpower expectations and policy implications, by G. Nerb (March 1986).
- No. 45 Le taux de profit et ses composantes dans l'industrie française de 1959 à 1981, par Angelo Reati (Mars 1986).
- No. 46 Forecasting aggregate demand components with opinions surveys in the four main EC-Countries Experience with the BUSY model, by M. Biart and P. Praet (May 1986).
- No. 47 Report of CEPS Macroeconomic Policy Group : Reducing Unemployment in Europe : The Role of Capital Formation, by F. Modigliani, M. Monti, J. Drèze, H. Giersch and R. Layard (July 1986).
- No. 48 Evolution et problèmes structurels de l'économie française, par X. Lannes, B. Philippe et P. Lenain (août 1986).
- No. 49 Long run implications of the increase in taxation and public debt for employment and economic growth in Europe by G. Tullio (August 1986).
- No. 50 Consumers Expectations and Aggregate Personal Savings by Daniel Weiserbs and Peter Simmons (November 1986).
- No. 51 Do after tax interest affect private consumption and savings ? Empirical evidence for 8 industrial countries : 1970-1983 by G. Tullio and Fr. Contesso (December 1986).
- No. 52 Validity and limits of applied exchange rate models : a brief survey of some recent contributions by G. Tullio (December 1986).

- No. 53 Monetary and Exchange Rate Policies for International Financial Stability : a Proposal by Ronald I. McKinnon (November 1986).
- No. 54 Internal and External Liberalisation for Faster Growth by Herbert Giersch (February 1987).
- No. 55 Regulation or Deregulation of the Labour Market : Policy Regimes for the Recruitment and Dismissal of Employees in the Industrialised Countries by Michael Emerson (June 1987).
- No. 56 Causes of the development of the private ECU and the behaviour of its interest rates : October 1982 - September 1985 by G. Tullio and Fr. Contesso (July 1987).
- No. 57 Capital/Labour substitution and its impact on employment by Fabienne Ilzkovitz (September 1987).
- No. 58 The Determinants of the German Official Discount Rate and of Liquidity Ratios during the classical gold standard : 1876-1913 by Andrea Sommariva and Giuseppe Tullio (September 1987).
- No. 59 Profitability, real interest rates and fiscal crowding out in the OECD area 1960-1985 (An examination of the crowding out hypothesis within a portfolio model) by Jorgen Mortensen (October 1987).
- No. 60 The two-handed growth strategy for Europe : Autonomy through flexible cooperation by J. Drèze, Ch. Wyplosz, Ch. Bean, Fr. Giavazzi and H. Giersch (October 1987).
- No. 61 Collusive Behaviour, R & D, and European Policy by Alexis Jacquemin (November 1987).
- No. 62 Inflation adjusted government budget deficits and their impact on the business cycle : empirical evidence for 8 industrial countries by G. Tullio (November 1987).
- No. 63 Monetary Policy Coordination Within the EMS : Is There a Rule ? by M. Russo and G. Tullio (April 1988).
- No. 64 Le Découplage de la Finance et de l'Economie Contribution à l'Evaluation des Enjeux Européens dans la Révolution du Système Financier International par J.-Y. Haberer (Mai 1988).
- No. 65 The completion of the internal market : results of macroeconomic model simulations by M. Catinat, E. Donni and A. Italianer (September 1988).
- No. 66 Europe after the crash : economic policy in an era of adjustment by Charles Bean (September 1988).
- No. 67 A Survey of the Economies of Scale by Cliff Pratten (October 1988).
- No. 68 Economies of Scale and Intra-Community trade by Joachim Schwalbach (October 1988).
- No. 69 Economies of Scale and the Integration of the European Economy : the Case of Italy by Rodolfo Helg and Pippo Ranci (October 1988).
- No. 70 The Costs of Non-Europe An assessment based on a formal Model of Imperfect Competition and Economies of Scale by A. Smith and A. Venables (October 1988).
- No. 71 Competition and Innovation by P.A. Geroski (October 1988).
- No. 72 Commerce Intra-Branche Performances des firmes et analyse des échanges commerciaux dans la Communauté européenne par le Centre d'Etudes Prospectives et d'Informations Internationales de Paris (octobre 1988).
- No. 73 Partial Equilibrium Calculations of the Impact of Internal Market Barriers in the European Community by Richard Cawley and Michael Davenport (October 1988).
- No. 74 The exchange-rate question in Europe by Francesco Giavazzi (January 1989).
- No. 75 The QUEST model (Version 1988) by Peter Bekx, Anne Bucher, Alexander Italianer, Matthias Mors (March 1989).
- No. 76 Europe's Prospects for the 1990s by Herbert Giersch (May 1989)
- No. 77 1992, Hype or Hope : A review by Alexander Italianer (February 1990)
- No. 78 European labour markets : a long run view (CEPS Macroeconomic Policy Group 1989 Annual Report) by J.-P. Danthine, Ch. Bean, P. Bernholz and E. Malinvaud (February 1990)
- No. 79 Country Studies The United Kingdom by Tassos Belessiotis and Ralph Wilkinson (July 1990)
- No. 80 See "Länderstudien" No. 1
- No. 81 Country Studies The Netherlands by Filip Keereman, Françoise Moreau and Cyriel Vanbelle (July 1990)

- No. 82 Country Studies Belgium by Johan Baras, Filip Keereman and Françoise Moreau (July 1990)
- No. 83 Completion of the internal market : An application of Public Choice Theory by Manfred Teutemann (August 1990)
- No. 84 Monetary and Fiscal Rules for Public Debt Sustainability by Marco Buti (September 1990)
- No. 85 Are we at the beginning of a new long term expansion induced by technological change?, by AngeloReati (August 1991)
- No. 86 Labour Mobility, Fiscal Solidarity and the Exchange Rate Regime: a Parable of European Union and Cohesion, by Jorge Braga de Macedo (October 1991)
- No. 87 The Economics of Policies to Stabilize or Reduce Greenhouse Gas Emissions: the Case of CO₂, by Mathias Mors (October 1991)
- No. 88 The Adequacy and Allocation of World Savings, by Javier Santillán (December 1991)

Country studies*

See also Economic Papers No.79 (The United Kingdom), No.81 (The Netherlands) and No.82 (Belgium).

- No. 1 The Federal Republic of Germany (September 1990)
- No. 2 Portugal (February 1991)
- No. 3 United Kingdom (March 1991)
- No. 4 Denmark (April 1991)
- No. 5 France (août 1991)

^{*} Country studies are the result of internal analysis of the economic situation of the respective member country; they are made on the responsibility of the Directorate-General for Economic and Financial Affairs of the Commission of the European Communities.